Author's response to reviews

Title: Near-infrared (NIR) spectroscopy. A new method for arthroscopic evaluation of low grade degenerated cartilage lesions. Results of a pilot study

Authors:

Gunter Spahn (spahn@pk-eisenach.de)
Holger Plettenberg (hplettenberg@fzmb.de)
Enriko Kahl (e.kahl@med.uni-goettingen.de)
Hans M Klinger (michael.klinger@med.uni-goettingen.de)
Thomas Muckley (Thomas.Mueckley@med.uni-jena.de)
Gunther O Hofmann (gunther.hofmann@med.uni-jena.de)

Version: 4 Date: 26 May 2007

Author's response to reviews: see over
Editorial managing team,

The manuscript has undergone newly formatting revision. Accordingly to your suggestions same corrections were done.

Please remove the authors' qualifications/titles (Dr.) from the title page.

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

→ done.

The emails should all be listed together be separate from the author list.

→ done.

Affiliations - Each affiliation should be written in full and be provided in the following format: department, institute, city, country. Please do not use full capitals for country names, and please do not include full-stops in abbreviated country names.

→ done.

Can we also ask you to link each author to their corresponding affiliation with the use of superscript numbers.

→ done.

For example:

Please mark the corresponding author with a symbol.

→ done.

Please remove the line numbers and the repeated title (above the abstract) from the manuscript.

→ done.

Figure 1 has been uploaded as 4 separate files. Please combine these into one file and ensure that the parts are labelled as "(a)", "(b)", "(c)" and "(d)" only. Alternatively you can leave the files as they are and update your figure citations to incorporate all 6 uploaded files. This means that your "Figure 2" is in fact "Figure 5" and your "Figure 3" is in fact "Figure 6".

→ done.

Table 2 has not been cited in the main text. Please ensure that all tables and figures are mentioned in the text in a numerically consecutive order starting with Table 1 and Figure 1.

→ done.

Table 2 has not been cited in the main text. Please ensure that all tables and figures are mentioned in the text in a numerically consecutive order starting with Table 1 and Figure 1.

→ done.

Tables. We are unable to display shading and colours within the tables. Can we therefore ask you to use another method of highlighting the shaded text e.g. by making it bold.

→ done.

Please remove the additional files from the submission system as they have already been included correctly in the manuscript.

→ done.
Thank you for assistance in making these corrections.

Best regards

G. Spahn

Dr. med. Gunter Spahn
Center of Trauma and Orthopaedic Surgery Eisenach
Sophienstr. 16, 99817 Eisenach, Germany
Tel:  ++49 (0) 3691 73 500
Fax:  ++49 (0) 2691 73 5011
Email: spahn@pk-eisenach.de